Author's response to reviews

Title: Functional analyses of ATM, ATR and Fanconi anemia proteins in lung carcinoma

Authors:

Jan H Beumer (beumerjh@upmc.edu)
Katherine Y Fu (kayfu01@gmail.com)
Bean N Anyang (njohbean@gmail.com)
Jill M Siegfried (jsiegfri@umn.edu)
Christopher J Bakkenist (bakkenistcj@upmc.edu)

Version: 6
Date: 12 August 2015

Author's response to reviews:

Figure 1 is relabelled as Figure 1.
The Introduction is renamed to Background.